

G OPEN ACCESS

Citation: Willen L, Basáñez M-G, Dvorak V, Veriegh FBD, Aboagye FT, Idun B, et al. (2021) Human immune response against salivary antigens of *Simulium damnosum* s.l.: A new epidemiological marker for exposure to blackfly bites in onchocerciasis endemic areas. PLoS Negl Trop Dis 15(6): e0009512. <u>https://doi.org/10.1371/journal.</u> pntd.0009512

Editor: Samuel Wanji, University of Buea, CAMEROON

Received: December 2, 2020

Accepted: May 27, 2021

Published: June 22, 2021

Copyright: © 2021 Willen et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its <u>Supporting Information</u> files. The data supporting the conclusions of this article are included within the manuscript and its Supporting Information files. The mass spectrometry proteomics data have been deposited to the ProteomeXchange Consortium (<u>http://</u> <u>proteomecentral.proteomexchange.org</u>) via the PRIDE partner repository with the dataset identifier PXD025869 and DOI <u>10.6019/PXD025869</u>. RESEARCH ARTICLE

Human immune response against salivary antigens of *Simulium damnosum* s.l.: A new epidemiological marker for exposure to blackfly bites in onchocerciasis endemic areas

Laura Willen^{1*}, Maria-Gloria Basáñez², Vit Dvorak¹, Francis B. D. Veriegh³, Frank T. Aboagye³, Bright Idun³, Maha Elhadi Osman⁴, Mike Y. Osei-Atweneboana³, Orin Courtenay⁵, Petr Volf¹

1 Department of Parasitology, Faculty of Science, Charles University, Prague, Czech Republic, 2 MRC Centre for Global Infectious Disease Analysis and London Centre for Neglected Tropical Disease Research, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom, 3 Biomedical and Public Health Research Unit, CSIR-Water Research Institute, Accra, Ghana, 4 Commission for Biotechnology and Genetic Engineering, National Centre for Research, Khartoum, Sudan, 5 Zeeman Institute for Systems Biology & Infectious Disease Epidemiology Research and School of Life Sciences, University of Warwick, Coventry, United Kingdom

* laura.willen@gmail.com

Abstract

Background

Simulium damnosum sensu lato (s.l.) blackflies transmit *Onchocerca volvulus*, a filarial nematode that causes human onchocerciasis. Human landing catches (HLCs) is currently the sole method used to estimate blackfly biting rates but is labour-intensive and questionable on ethical grounds. A potential alternative is to measure host antibodies to vector saliva deposited during bloodfeeding. In this study, immunoassays to quantify human antibody responses to *S. damnosum* s.l. saliva were developed, and the salivary proteome of *S. damnosum* s.l. was investigated.

Methodology/Principal findings

Blood samples from people living in onchocerciasis-endemic areas in Ghana were collected during the wet season; samples from people living in Accra, a blackfly-free area, were considered negative controls and compared to samples from blackfly-free locations in Sudan. Blackflies were collected by HLCs and dissected to extract their salivary glands. An ELISA measuring anti-*S. damnosum* s.l. salivary IgG and IgM was optimized and used to quantify the humoral immune response of 958 individuals. Both immunoassays differentiated negative controls from endemic participants. Salivary proteins were separated by gel-electrophoresis, and antigenic proteins visualized by immunoblot. Liquid chromatography mass spectrometry (LC–MS/MS) was performed to characterize the proteome of *S. damnosum* s. I. salivary glands. Several antigenic proteins were recognized, with the major ones located around 15 and 40 kDa. LC–MS/MS identified the presence of antigen 5-related protein, apyrase/nucleotidase, and hyaluronidase.

Funding: The work was supported by the Global Challenges Research Fund (Networks in Vector Borne Disease Research Gnatwork) BBSRC grant (BB/R005362/1) to OC, MGB, PV, MYO-A and MO. LW and PV are further supported by the Ministry of Education of the Czech Republic through the European Regional Development Fund (projects "BIOCEV – Biotechnology and Biomedicine Centre of the Academy of Sciences and Charles University" CZ.1.05/1.1.00/02.0109 and "CePaViP", CZ.02.1.01/0.0/0.0/16 019/0000759). MGB acknowledges joint centre funding (grant No. MR/ R015600/1) by the UK Medical Research Council (MRC) and the UK Department for International Development (DFID) under the MRC/DFID Concordat agreement, which is also part of the **European and Developing Countries Clinical Trials** Partnership (EDCTP2) programme supported by the European Union. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions/Significance

This study validated for the first time human immunoassays that quantify humoral immune responses as potential markers of exposure to blackfly bites. These assays have the potential to facilitate understanding patterns of exposure as well as evaluating the impact of vector control on biting rates. Future studies need to investigate seasonal fluctuations of these antibody responses, potential cross-reactions with other bloodsucking arthropods, and thoroughly identify the most immunogenic proteins.

Author summary

Measuring biting rates of bloodsucking arthropods has been proved to be a crucial tool in understanding transmission dynamics and evaluating control measures against vectorborne diseases. Blackflies of the *S. damnosum sensu lato* (s.l.) complex are the prime vectors of *Onchocerca volvulus*, a filarial nematode and causative agent of human onchocerciasis (or river blindness) in Africa. Blackfly biting rates are typically measured by human attractant vector collectors who capture host-seeking female flies as they alight on their skin but before they take a blood meal. Apart from the ethical concerns raised by this method, estimates thus obtained do not reflect true exposure to vector bites. An understanding of whether there are people who receive more bites than others and who may be at higher risk of acquiring the infection is crucial to modelling control interventions. Therefore, in the present study we developed IgG- and IgM-based immunoassays to quantify antibody responses in human populations to salivary antigens of *S. damnosum* s.l. in onchocerciasis-endemic communities of Ghana. The proteome of *S. damnosum* s.l. salivary glands was also investigated to potentially identify salivary antigens that could be employed as exposure markers against this important vector group.

Introduction

Human onchocerciasis or river blindness is a severely debilitating parasitic disease caused by the filarial nematode *Onchocerca volvulus* (Nematoda: Filarioidea) and is transmitted via bites of infective haematophagous female blackflies (Diptera: Simuliidae). Although the disease has been endemic in six Latin-American countries and still occurs in Yemen, 99% of the cases are found in Africa where the most important vectors belong to the *Simulium damnosum sensu lato* (s.l.) species complex [1]. The sibling species within this complex vary in their ecology and geographical distribution, and exhibit different degrees of anthropophagy, vector competence, and vectorial capacity [2, 3].

The World Health Organization's 2021–2030 Roadmap for Neglected Tropical Diseases has proposed that elimination (interruption) of transmission of onchocerciasis be verified in an increasing number of countries by 2030 [4]. Transmission dynamics models helped to inform the formulation of this roadmap, and indicated that vector control (where possible, and in addition to regular treatment of human populations) would be favourable in reducing blackfly biting rates, and hence transmission, in highly endemic areas [5]. An important determinant of the feasibility of elimination is the baseline (pre-intervention) level of endemicity, defined by the initial infection prevalence. Onchocerciasis prevalence exhibits a strongly non-linear relationship with the annual biting rate (ABR, the number of blackfly bites per person per

year) [6], such that high prevalence values can be associated with a broad range of ABRs [7]. Crucial modelling uncertainties include the assumed degree of exposure heterogeneity in the human population, and the age- and sex-specific patterns of exposure [7, 8]. Stronger heterogeneity (ultimately resulting in few hosts harbouring most parasites) hinders elimination [9]. However, measuring ABRs is labourious, time-consuming and mostly reliant on human landing catches (HLCs) [10]. Measuring individual heterogeneity in exposure to blackfly bites requires counting the actual number of blackflies landing and successfully feeding on individual villagers during their routine day-time activities [11]. These procedures raise ethical concerns about enhanced risk of infection for those involved and are unable to provide robust measures of effective exposure to biting between and within endemic communities. Therefore, developing and validating alternative methods for measuring exposure to blackfly bites is highly desirable. Data obtained using such methods would improve the parameterization of onchocerciasis transmission models by using independent exposure data, since currently exposure patterns are inferred from parasitological data, which compound exposure to vector bites and infection processes [7]. At the same time, such methods would provide invaluable tools to evaluate and compare the impact of various anti-vectorial approaches for onchocerciasis control and elimination [12-14]. Developing assays to quantify anti-blackfly's saliva antibody responses in endemic human populations would potentially provide such tools, as demonstrated with sand flies and leishmaniasis [15, 16].

Saliva of haematophagous arthropods is known to contain a vast array of pharmacologically-active compounds with anti-haemostatic (i.e. vasodilators, inhibitors of blood coagulation and platelet aggregation), anti-inflammatory and immunomodulatory activities (reviewed in [17]). Besides the pharmacological function of many of these compounds, studies on mosquitoes, triatomines, tsetse flies, and sand flies have proved that their salivary proteins provide useful tools to develop immunoassays to estimate host exposure to their bites [16, 18–21]. In blackflies, salivary components of relatively few species have been researched; in fact, immunomodulatory activities and (murine) humoral immune responses against salivary antigens have only been identified for *S. vittatum* [22, 23]. The saliva injected when blackflies bite causes a pronounced and persistent erythema, together with intense itching and local swelling. Previous studies have demonstrated the occurrence of IgE-mediated allergic dermatitis in horses induced by *S. vittatum* saliva [24] and IgE-mediated hypersensitivity in humans who exhibited allergic reactions to the bite of *S. nigrogilvum* [25]. However, no assays have been developed to measure human exposure to simuliid vectors of human onchocerciasis.

In this study, we examine the humoral immune response against S. damnosum s.l. in humans living in onchocerciasis-endemic communities in Ghana and propose a novel tool to evaluate blackfly exposure in onchocerciasis epidemiological and vector control studies. Ghana was one of the first countries to initiate vector control activities by large-scale aerial larviciding of blackfly breeding sites under the former Onchocerciasis Control Programme in West Africa (OCP, 1974–2002) [26]. It was also one of the first countries to commence mass drug administration (MDA) to complement vector control after the completion of a 5-year community trial of ivermectin (Mectizan), conducted in the former hyperendemic focus of Asubende [27]. Currently, ivermectin MDA is the mainstay of onchocerciasis control and elimination in Africa (including Ghana) as large-scale deployment of blackfly control is no longer performed. The focus of Asubende has been the subject of comprehensive entomological research in relation to onchocerciasis transmission [28-31], and became a source of epidemiological and entomological data for parameterization of transmission models [3, 32, 33]. In this paper, we report on the development of immunoassays against S. damnosum s.l. salivary antigens, provide a characterization of its proteome, and discuss the potential use of certain salivary components as host exposure markers, the results of which can be used to improve

our understanding of onchocerciasis transmission dynamics towards its control and elimination in African settings.

Materials and methods

Ethics statement

Ethical clearance was obtained from the Council for Scientific and Industrial Research (CSIR) Institutional Review Board (RPN008/CSIR-IRB/2019). Prior to the study, all participants were informed in detail in their local language about the purpose of the study. Fully informed signed consent was subsequently obtained from the participant, or from the parent/ guardian in case the participant was under 18 years of age. All participants were registered before blood sampling. Data were anonymized. Participating individuals were not at an increased risk of exposure. All participants were living in an area that is part of the national MDA programme through which they receive biannual ivermectin treatment [34, 35]. Blackflies were collected by OCP vector collectors following standard OCP vector collector techniques [10]. Archived Sudanese human plasma samples used to validate the assays presented in this paper were obtained under ethical approval as previously described [36].

Blood samples

Blood samples of 958 people living in rural villages in an onchocerciasis endemic area in the Bono East region (a new region, created from partitioning the former, larger Brong-Ahafo region threefold) of Ghana were collected during the wet season in August 2019. The area is reported to have high S. damnosum s.l. biting rates as determined by standard OCP vector collector techniques [29, 37]. Blood samples were obtained from village residents over the age of 5 years, with a target of at least 100 per age/sex class (age categories in years: 5-10, 11-20, 21-30, 31-40, 41-50, 51-60, 61-70, >71) by adopting two approaches depending on the population size of the village. In villages where the numbers within age/sex categories were generally small, all participants were offered the possibility of recruitment; in larger villages an age/sex stratified sample was identified based on inspection of the population medical register of MDA treatment contained in three books, in which the family names of all villagers were recorded. Within-book page numbers were selected by a random number generator in R software, and the corresponding families were invited to participate. This procedure was repeated until the target numbers within age/sex classes was reached in equal numbers as far as logistically possible. Blood samples from the OCP vector collectors living in the same area and expected to have experienced high biting rates were also collected and served as positive control samples during the preparation of the immunoassays. Thirty-five people at the Water Research Institute (CSIR) in Accra aged between 30 and 59 years of age and no known history of travelling to blackfly/ onchocerciasis-endemic areas were used as non-simuliid bitten negative controls. Twenty samples from people living in blackfly-free rural communities in Gedaref state and the city of Khartoum, Sudan, were used to further validate the negative control plasma samples from Accra. From each recruit 2 ml of blood was collected, placed in a cool box for transport, and spun down at $2,000 \times g$ for 15 min to transfer the plasma to a separate tube which was stored at -20° C until testing.

Blackfly sample collection and identification

In Bono East, blackfly biting rates have previously been recorded in the dry season (2,061 bites/person/month in Asubende; 775 in Agborlekame [29]; 500–2,500 bites/person/month

in Asubende/Senyase and Aborlekame/Tainso [<u>37</u>], with only *S. damnosum sensu stricto* (s.s.)/*S. sirbanum* (the savannah members of the *S. damnosum* s.l. complex) being identified along the Pru river [<u>28</u>, <u>29</u>]. In other regions of Ghana where dry and wet season sampling has been conducted, biting rates were higher in the wet season [<u>29</u>]. Therefore, for this study, sampling was performed in the wet season, the period with anticipated higher biting rates and plausibly also higher antibody responses.

Host-seeking blackflies were collected along the Pru river near one of the study villages (Asubende; $08^{\circ} 01' 01.4"$ N, $00^{\circ} 58' 53.8"$ W) by two OCP vector collectors using standard HLC techniques [10]. Blackflies were collected for 15 consecutive days in August 2019, from 7 a.m. until 6 p.m. or until the daily target of 200 blackfly females was reached. Pairs of workers collected flies in hourly intervals alternating between themselves. Collectors sat with bare lower legs and arms to capture landing female blackflies individually in polypropylene tubes (7.5 cm × 1.2 cm, Sigma-Aldrich), and immediately on landing before the flies took a blood meal. The blackflies were placed in a cool box for transport until dissection later the same day.

A previous study conducted in the same area in Ghana demonstrated that all female flies caught by vector collectors were within the *S. damnosum* s.l. species complex [29]. To confirm this, a subset of the caught specimens (n = 18) were stored in 96% ethanol, their total genomic DNA extracted by High Pure PCR Template Preparation Kit (Roche Life Science, Penzberg, Germany), and a fragment of mitochondrial *ND4* gene amplified using the ND4-1/ND4-2 primer pair using reaction conditions as described previously [38]. PCR products were purified using a QIAquick PCR Purification Kit (Qiagen, Hilden, Germany) and directly sequenced in both directions using the same amplification primers. The obtained nucleotide sequences were edited and aligned using BioEdit software (7.0.9.0) [39] and compared by BLAST (basic local alignment search tool) with those available in GenBank.

Dissection of blackfly salivary glands and preparation of the antigen

A total of 940 blackfly salivary glands were extracted by dissection under a stereomicroscope, immediately after the flies were anesthetized by placing them in a -20° C freezer for 10 min. Salivary glands were dissected in aliquots of 20 glands, stored in Tris-buffered saline (TBS, pH 7.5) at a concentration of one salivary gland per μ l TBS and kept in a cool box during dissections. The aliquots were stored at -20° C until preparation of the assays.

Prior to the immunoassay and western blot analyses, the salivary glands were disrupted by three quick freeze-and-thaw cycles in liquid nitrogen. The aliquots of the salivary glands homogenates (SGH) were then spun at 13,500 rpm for 5 min to pellet any debris, and the supernatant of all aliquots was collected and pooled together. The concentration of the final pool was approximated by a nanodrop ND-1000 spectrophotometer (ThermoFisher Scientific) at 280 nm and was \pm 0.94–1.02 mg/ml. The pool of SGH was aliquoted and used for both the enzyme-linked immunosorbent assays (ELISA) and the western blot.

Enzyme-linked immunosorbent assays

Optimal antigen concentration and plasma dilution to measure anti-*S. damnosum* s.l. IgG were identified during an optimization process of which details are provided in Text A in <u>S1 File</u>.

All 958 plasma samples were then tested by ELISA for IgG, and a subset of 500 samples selected by random number generation from the full sample set within stratified age-classes (target 65 per age-class) was tested for IgM, based on the optimal dilutions defined during

the preliminary assays. For the IgG and IgM assays, 96-well microtiter plates were coated, respectively, with 0.2 µg and 0.025 µg of SGH in 100 µl of 20 mM carbonate-bicarbonate coating buffer (pH 9.5) per well. After overnight incubation at 4° C, the plates were washed twice with phosphate-buffered saline with 0.05% Tween 20 (PBS-Tw), and blocked with 6% non-fat dried milk (Bio-Rad) diluted in PBS-Tw for 1 h at 37° C. After washing twice with PBS-Tw, the plates were incubated with plasma diluted at 1/100 (IgG) or 1/50 (IgM) in 2% non-fat dried milk and incubated for 1.5 h at 37° C. Plates were subsequently washed 6 times with PBS-Tw and incubated for 45 min at 37° C with peroxidase-conjugated anti-human IgG or IgM antibody (Sigma-Aldrich; Bethyl Laboratories, Inc.) diluted at 1/1,000 and 1/ 100,000 in PBS-Tw, respectively. The plates were washed 6 times and the chromogenic reaction was developed in the dark using an orthophenylendiamine (OPD) solution in a phosphate-citrate buffer (pH 5.5) with 0.1% hydrogen peroxide. The reaction was stopped after 5 min with 10% sulfuric acid and the absorbance (OD value) was measured at 492 nm using a Tecan Infinite M200 microplate reader (Schoeller). In each step, 100 µl of solution was used per well.

Standardization

All samples were tested in duplicate. Samples with a coefficient of variation of more than 20% were retested. Each plate included a blank control well of which the OD was subtracted from the sample OD values for preliminary interrogation of the results. In order to correct for interplate variability, a set of positive control (PC) and negative control (NC) samples was included in each plate and standardized OD values (SOD) were generated according to the following formula: SOD = OD _{sample}/(Average OD_{PC}-average OD_{NC}).

Statistical analysis

All statistical analyses were performed using R software [40]. The median OD values between the optimization assays and the median SOD values between the different groups in the final assay were compared using the Wilcoxon signed rank sum test with Holm correction. A *P*-value of <0.05 was considered statistically significant. Cut-off values for the final assays were calculated based on the mean of the negative control samples plus three standard deviations. The results were graphically presented using the "ggplot2" package in R software [41].

Sodium dodecyl sulphate-polyacrylamide gel electrophoresis (SDS-PAGE) and western blot

The salivary proteins present in blackfly SGH were electrophoretically separated on a 12% polyacrylamide gel under non-reducing conditions using a Mini-protean apparatus (Bio-Rad) (200 V, 55 min). A total of 4 μ g of SGH was loaded per lane. One lane containing the blackfly separated salivary proteins was silver-stained, the rest was blotted onto the nitro-cellulose membrane using the iBLOT instrument (Invitrogen), according to the manufacturer's instructions. The nitrocellulose membrane was cut into strips and blocked at 4° C overnight in 6% milk diluted in TBS with 0.05% Tween 20 (TBS-Tw). The immunoblot was carried out with five negative control samples and 20 endemic plasma samples stratified by the IgG SOD values of <0.5 (group 1), 0.5–1.5 (group 2), and >1.5 (group 3). The nitrocellulose strips were then incubated with the individual samples diluted at 1:100 in TBS-Tw. After 1 h, all strips were washed with TBS-Tw and incubated for 1 h with anti-human IgG HRP conjugated antibody (1:1,000) (Sigma-Aldrich). Finally, the chromogenic reaction

was developed with a substrate solution containing 3,3'-diaminobenzidine (Sigma-Aldrich) and H_2O_2 .

Protein characterization by liquid chromatography-tandem mass spectrometry

To characterize the proteins by liquid chromatography-tandem mass spectrometry (LC-MS/ MS) analysis via in-gel digestion, blackfly SGH was electrophoretically separated following the same protocol as used for the Western Blot and stained with Coomassie G-250. Individual bands containing the proteins of interest were then excised from the Coomassie G-250 stained SDS-PAGE gel using sterile scalpels and cut into small pieces (approx. 1 mm × 1 mm). Bands were de-stained by sonication for 30 min in 50% acetonitrile (ACN) and 50 mM ammonium bicarbonate (ABC). After de-staining, the solution was removed and the gels were dried in ACN. Disulfide bonds were reduced using 10 mm dithiothreitol (DTT) in 100 mM ABC at 60° C for 30 min. Samples were again dried with ACN and free cysteine residues were blocked at room temperature in the dark for 10 min using 55 mM iodoacetamide in 100 mM ABC. Samples were dried thoroughly and the digestion buffer (10% ACN, 40 mM ABC and 13 ng/µl trypsin) was added to cover the gel pieces. Proteins were digested at 37° C overnight, after which they were sonicated for 30 min using 150 µl of 50% ACN with 0.5% formic acid. The supernatant containing the peptides was transferred to a new microcentrifuge tube to which 150 µl of elution solution was added. Following a 30 min sonication, the solution was removed, combined with the previous solution and dried using Speedvac. The dried peptides were reconstituted in 2% ACN with 0.1% trifluoroacetic acid (TFA) and injected into the Ultimate 3000 Nano LC coupled to the Orbitrap Fusion.

Nano Reversed phase column (EASY-Spray column, 50 cm \times 75 μ m ID, PepMap C18, 2 µm particles, 100 Å pore size) was used for LC-MS/MS analysis. Mobile phase buffer A was composed of water and 0.1% formic acid. Mobile phase B was composed of ACN and 0.1% formic acid. Samples were loaded onto the trap column (Acclaim PepMap300, C18, 5 µm, 300 Å Wide Pore, 300 μ m \times 5 mm) at a flow rate of 15 μ l/min. The loading buffer was composed of water, 2% ACN and 0.1% TFA. Peptides were eluted with a gradient of B going from 4% to 35% over 60 min at a flow rate of 300 nl/min. The eluted peptide cations were converted to gas-phase ions by electrospray ionization and analyzed on a Thermo Orbitrap Fusion (Q-OTqIT, Thermo). Survey scans of the peptide precursors from 350 to 1,400 m/z were performed at 120 K resolution (at 200 m/z) with a 5×10^5 ion count target. Tandem MS was performed by isolation at 1.5 Th with the quadrupole, higher energy collisional dissociation (HCD) fragmentation with normalized collision energy of 30, and rapid scan MS analysis in the ion trap. The MS/MS ion count target was set to 10⁴ and the maximum injection time was 35 ms. Only those precursors with charge state 2-6 were sampled for MS/MS. The dynamic exclusion duration was set to 45 s with a 10 ppm tolerance around the selected precursor and its isotopes. The monoisotopic precursor selection was turned on. The instrument was run in top speed mode with 2 s cycles [42].

All data were analyzed and quantified with the MaxQuant software (version 1.6.1.0) [43]. The false discovery rate (FDR) was set to 1% for both proteins and peptides and a minimum length of seven amino acids was specified. The Andromeda search engine was used to screen the MS/MS spectra against the Simuliidae database (downloaded from NCBI in June 2020, containing 16,564 entries), and the *Culicoides nubeculosus* (downloaded from Uniprot in May 2020, containing 79 entries) and *Culicoides sonorensis* databases (downloaded from Uniprot in March 2021, containing 21,090 entries) (Ceratopogonidae). Enzyme specificity was set as C-terminal to Arg and Lys allowing cleavage at proline bonds and a maximum of two missed

cleavages. Carbamydomethylation of cysteines was selected as a fixed modification and N-terminal protein acetylation, methionine oxidation and serine/threonine phosphorylation were selected as variable modifications. Data analysis was performed using Perseus 1.5.2.4 software [44].

Results

ELISA optimization

The optimal dilutions for the SGH, the anti-human IgG and IgM conjugates and the plasma samples were determined during preliminary assays. Details of the optimization process are provided in Text A and Figs A and B and Table A in <u>S1 File</u>.

Human IgG and IgM immunological responses

All endemic Bono East human samples collected during the study (n = 958) and a subset of 500 samples were tested in duplicate by the IgG and IgM ELISA, respectively, together with negative control samples (n = 35 and n = 33 for the IgG and IgM ELISA, respectively). According to the cut-off values, 48.4% of the participants was positive for IgM, whereas 99.5% tested positive for IgG. The distributions of the SOD values for both ELISAs are shown in Fig 1. A statistically significant difference was detected between the median IgG SOD values of the Bono East test samples (median = 1.35 [quartile Q1: 1.14, quartile Q3: 1.49]) and the negative control plasma samples (median = 0.0987 [Q1: 0.08, Q3: 0.13]) (P<0.001). Median IgM SOD values for the two groups were also significantly different (median = 0.975 [Q1: 0.75, Q3: 1.36] and 0.384 [Q1: 0.28, Q3: 0.53]), respectively (P<0.001).

Lastly, 20 plasma samples from individuals living in blackfly-free rural communities in Gedaref state and the city of Khartoum, Sudan, were used to validate the control samples from Accra in the IgG ELISA. The distributions of the IgG SOD values for both negative controls groups were similar and not significantly different (Fig C in <u>S1 File</u>).

Immunoblot with blackfly salivary proteins

Simulium damnosum s.l. IgG immunogenic proteins were identified by immunoblotting (Fig 2). A total of 20 plasma samples of individuals from the Bono East region were tested together with five negative control samples from people living in Accra. The positive samples were chosen according to their SOD value in the anti-S. damnosum s.l. IgG ELISA, representing SOD values of <0.5 (group 1), 0.5–1.5 (group 2), and >1.5 (group 3). No protein bands were recognized in samples from Accra, indicating that non-specific binding of human plasma to blackfly salivary antigens is unlikely. In contrast, samples from the endemic Bono East region recognized several protein bands with different intensities. People with an SOD value >0.5 recognized several proteins in the range of 20–40 kDa. Most individuals recognized a protein with an apparent molecular weight (MW) of 17 kDa, two proteins with an approximate MW of 38 kDa, and two proteins with MWs around 60-70 kDa. One major band around 15 kDa was recognized by all samples. The intensity of protein recognition decreased with decreasing SOD values and, as expected, individuals with lower SOD values in the ELISA recognized less protein bands. Interestingly, five samples in the group with an SOD value < 0.5, classified as negative according to the ELISA cut-off value, showed strong reactivity with some major protein salivary bands around 15 and 38 kDa, and lower reactivity with bands around 17 and 60-70 kDa.



Fig 1. Distribution of standardized optical density (SOD) values for the IgG and IgM ELISA. A. SOD values of the IgG ELISA for negative control samples (n = 35) from Accra and Bono East region residents (n = 958), Ghana. The cut-off value is shown by a dashed horizontal line and is equal to 0.275 SOD. B. SOD values of the IgM ELISA for negative control

samples (n = 33) and Bono East region residents (n = 500). The cut-off value is shown by a dashed horizontal line and is equal to 0.998 SOD. The solid black horizontal line within the boxes is the median; the lower and upper borders are, respectively, the 1st (Q1) and 3rd (Q3) quartiles; the vertical bars indicate the 'minimum' and 'maximum' values, calculated as $Q1-1.5 \times IQR$ (interquartile range) and $Q1 + 1.5 \times IQR$, respectively.

https://doi.org/10.1371/journal.pntd.0009512.g001

Liquid chromatography-tandem mass spectrometry (LC-MS/MS) analysis

Based on the results from the immunoblot, several parts of a Coomassie G-250-stained SDS-PAGE gel were cut and analyzed by LC-MS/MS. The different regions that were cut from the gel are indicated on the Coomassie G-250-stained gel in Fig.3, showing that the separated proteins of *S. damnosum* s.l. saliva ranged between 10 and 80 kDa. A silver-stained gel was added to better visualize the separated proteins.

Proteins identified by LC-MS/MS are summarized in <u>Table 1</u>. Antigen 5-related proteins and major insect-specific and *Simulium*-specific protein families, enzymes, and protease inhibitors were detected. Antigens with a low MW were not specifically identifiable by mass-spectrometry (gel region 6–8). Apyrase/nucleotidase had the highest intensity in gel regions 1 and 3, salivary hyaluronidase in gel region 2, and salivary secreted antigen 5-related protein in gel regions 4 and 5.

Molecular identification of blackflies

Sequences obtained from the 18 analyzed simuliid specimens showed an identity of 99.5–100% with haplotypes of the *S. damnosum* complex [45], demonstrating that they all belong to *S. damnosum* s.l.



Fig 2. Immunoblot of salivary gland homogenate (SGH) of *Simulium damnosum sensu lato.* Five negative plasma samples from people living in Accra (Neg); nine plasma samples from people with an SOD value in the IgG ELISA greater than 1.5 (>1.5 SOD); six samples from people with an SOD value between 0.5 and 1.5 (0.5–1.5 SOD), and 5 samples from people with an SOD value lower than 0.5 (<0.5 SOD) were tested by western blot against SGH of S. damnosum s.l. STD: standard; SOD: Standardized Optical Density.

https://doi.org/10.1371/journal.pntd.0009512.g002





https://doi.org/10.1371/journal.pntd.0009512.g003

Discussion

Implications for measuring host exposure to simuliid bites

Studies of hosts bitten by bloodsucking arthropods have demonstrated a specific quantifiable humoral immune response that can potentially serve as an estimate of host exposure to bites of bloodsucking arthropods [16, 18–21]. Since no such assay existed for blackflies, we developed for the first time a human exposure assay for the important onchocerciasis vector group *S. damnosum* s.l., allowing us to examine humoral immune responses to salivary proteins by human populations in onchocerciasis (savannah) endemic areas in Ghana. We showed that specific anti-*S. damnosum* s.l. IgG and IgM responses are detectable and quantifiable, and we investigated the salivary antigenic proteins associated with these responses.

Previous studies measuring host exposure to salivary proteins of haematophagous arthropods mainly focused on the IgG response. Although IgG is an antibody that develops later after the initial antigen exposure, it has been shown to fluctuate throughout the season in case of sand flies and mosquitoes [46, 47]. This quality renders it ideal as an epidemiological tool to measure biting exposure in host populations. IgM responses, on the other hand, are the first to appear in response to an antigen, and with respect to arthropod saliva, have only been

Table 1. Proteome analysis of *S. damnosum* s.l. salivary proteins. Protein bands of interest were cut from a Coomassie G-250-stained SDS-PAGE gel and analyzed by liquid chromatography-tandem mass spectrometry (LC-MS/MS). The obtained data were screened against available cDNA libraries of Simuliidae, *Culicoides nubeculosus*, *and Culicoides sonorensis*. The identified proteins per gel region are listed according to their intensity with their GenBank accession number, the name of the protein, the species, and its molecular weight. Identified proteins highlighted in bold are the most intensely present per gel region.

Gel region	Matched peptides	Accession number	Protein name	Species	MW (kDa)	Intensity (x 1000)	Database
1 (>60kDa)	5	AEB96567.1	Putative apyrase/ nucleotidase, partial	S. guianense s.l.	34.339	702620	Simuliidae
	8	AEB96554.1	Salivary alpha-amylase, partial	S. guianense s.l.	61.471	177750	Simuliidae
	4	ACH56864.1	Salivary alpha-amylase, partial	S. vittatum	52.206	38972	Simuliidae
2 (45–60 kDa)	1	AEB96520.1	Salivary hyaluronidase	S. guianense s.l.	40.217	14988	Simuliidae
	1	ACZ28347.1	Putative apyrase/ nucleotidase, partial	S. nigrimanum	54.041	14973	Simuliidae
	4	AEB96567.1	Putative apyrase/ nucleotidase, partial	S. guianense s.l.	34.339	8517.3	Simuliidae
	3	AEB96562.1; ACZ28191.1; ACZ28190.1	Salivary alpha-amylase	S. guianense s.l., S. nigrimanum	46.021	2449.8	Simuliidae
3 (35–45 kDa)	3 3	AEB96567.1 ACZ28347.1; ACH56868.1	Putative apyrase/ nucleotidase, partial Putative apyrase/ nucleotidase, partial	S. guianense s.l. S. nigrimanum, S. vittatum	34.339 54.041	2613.7 2359.4	Simuliidae Simuliidae
4 (30–35 kDa)	1	ACH56844.1; ACH56843.1; ACZ28180.1; ACZ28179.1	Salivary secreted antigen-5 related protein	S. vittatum; S. nigrimanum	32.031	10754	Simuliidae
	2	AEB96567.1; ACH56868.1; ACZ28347.1	Putative apyrase/ nucleotidase, partial	S. guianense s.l., S. vittatum; S. nigrimanum	34.339	6338.1	Simuliidae
5 (24–35 kDa)	1	ACH56844.1; ACH56843.1; ACZ28180.1; ACZ28179.1	Salivary secreted antigen-5 related protein	S. vittatum, S. nigrimanum	32.031	440710	Simuliidae
	1	ACZ28347.1	Putative apyrase- nucleotidase, partial	S. nigrimanum	34.339	5689.3	Simuliidae

https://doi.org/10.1371/journal.pntd.0009512.t001

investigated as potential exposure markers of mosquito and triatomine bites [48, 49]. The early rise in specific IgM immune response makes it a potential candidate to evaluate the impact on bite exposure of recently implemented vector control methods, at both the population and individual level. Both IgG and IgM SOD values were substantially and statistically significantly higher in individuals residing in the Bono East region of Ghana than in their negative control counterparts, residing in blackfly-free areas of Ghana (Accra), supporting the notion that people living in onchocerciasis-endemic communities are exposed to *S. damnosum* s.l. bites. Although the Bono East region of Ghana has been part of Phase 1 (initiated in 1975) or the South-Eastern extension (started in 1986) of the OCP [30], larvicidal operations ended with the programme closure and fly densities returned to pre-control levels [29]. The lower IgM antibody responses might be indicative of its shorter half-life relative to that of IgG immune responses; however, blackfly biting occurs year-round [37], and the sampling was comprehensive across all age groups (5–95 years).

To further validate the IgG ELISA, samples from people living in blackfly-free rural communities in Gedaref state and in the city of Khartoum, Sudan, were tested showing similarly low and not significantly different baseline antibody levels to the Ghana control group, thus supporting the adopted cut-off values. Due to limitations in blackfly salivary gland availability, the Sudanese samples were not tested with the IgM ELISA. Notably, the antibody responses in the negative control group (in Ghana) of the proposed IgM ELISA were raised, resulting in a higher cut-off value. This may reflect a lower specificity of the IgM ELISA; skin hypersensitivity (IgE-mediated) cross-reactions between salivary allergens of *S. vittatum* and *C. nubeculosus* have been observed [50], but in sand flies and mosquitoes previous studies only reported cross-reactions between salivary proteins of closely-related species [21, 51, 52]. Evaluation of potential immunological cross-reactions between salivary proteins of *S. damnosum* s.l. and other bloodsucking arthropods would be informative to truly identify human exposure biomarkers unique for *S. damnosum* s.l..

Implications for unravelling the salivary proteome of *Simulium damnosum* s.l.

To date, the sialotranscriptomes of only three blackfly species have been unravelled. The first research group to uncover the sialotranscriptome of a member of the Simuliidae family was that of Andersen et al. [53], identifying several anti-haemostatic factors and immunomodulatory activities in *S. vittatum* saliva. Subsequent studies reported the sialotranscriptome of two other blackfly species, *S. nigrimanum* and *S. guianense* s.l. (the latter being a major onchocerciasis vector in the extant Amazonian focus straddling between Venezuela and Brazil [54]), contributing to the identification of additional compounds with potential anti-haemostatic or immunity-related functions [55, 56].

In the present study, antibodies of the Bono East residents recognized numerous *S. damnosum* s.l. salivary proteins. Importantly, people residing in Accra and who reported never to have travelled outside of the city, showed no evidence of any cross-reaction with *S. damnosum* s.l. salivary proteins, despite them undoubtedly having experienced bites from other haematophagous arthropods, including *Anopheles* mosquitoes [57]. People with lower SOD values reacted to fewer proteins and with lower intensity than people with higher SOD values. All five samples in the group with SOD values of less than 0.5 were classified as ELISA-negative yet showed strong reactivity with one protein band around 15 kDa; two of these five samples also reacted strongly with one band around 38 kDa, and (with a lower intensity) with a protein band around 17 kDa and two bands around 60–70 kDa. This finding could either indicate that the antibody response against these proteins is long-lived, making them a useful marker for past exposure, or that they are just more antigenic and thus potentially the best candidates to measure current exposure to *S. damnosum* s.l. bites.

Given the high immunogenicity of the *S. damnosum* s.l. salivary proteins, we attempted to characterize these proteins using mass-spectrometry analysis. Protein bands from *S. damno-sum* s.l. saliva separated by an SDS-PAGE Coomassie G-250-stained gel were thus subjected to LC-MS/MS and results were screened against the available cDNA libraries of Simullidae and of *C. nubeculosus* and *C. sonorensis*, as biting midges and blackflies are closely related and cross-reactions between salivary allergens of *S. vittatum* and *C. nubeculosus* have been observed [50]. Several well-known salivary proteins with known anti-haemostatic or immunity-related functions were found in the salivary gland proteome of *S. damnosum* s.l., including antigen 5-related protein, apyrase/ nucleotidase, and hyaluronidase. The strong antigens located at 15 kDa and around 17 kDa did not yield any major hits.

Antigen 5-related proteins are associated with IgE hypersensitivity responses against both *S. vittatum* and *S. nigrogilvum* [25, 58], with the *S. vittatum* protein sharing common IgEbinding epitopes with antigen 5-related proteins from *Culicoides nubeculosus* [50]. Apyrases/ nucleotidases are ubiquitously found in saliva of haematophagous arthropods. Although the origin of blackfly salivary apyrases has not yet been revealed, sialome studies of *S. vitattum*, *S. guianense* s.l., and *S. nigrimanum* all described transcripts coding for members of the 5'-nucleotidase family, similar to mosquito apyrases [53, 55, 56]. Interestingly, in blackflies apyrase expression was shown to be positively associated with anthropophagy and onchocerciasis vector status of the blackfly species [59]; a trend also observed for *Anopheles* vectors of malaria [60]. Allergenic and antigenic properties have been repeatedly reported for apyrases in mosquitoes and sand flies (reviewed in [61]). Lastly, hyaluronidase has mainly been demonstrated in the saliva of thelmophagous insects (i.e. pool-feeders), such as sand flies, biting midges, tabanid flies, and blackflies [62–65]. By cleaving hyaluronic acid (HA), the enzyme decreases tissue viscosity and consequently helps to enlarge the size of the feeding hematoma and spread other salivary components. It has also been shown to accelerate the bloodfeeding process and to enhance *Leishmania* transmission [65]. Similarly, it has been postulated that hyaluronidase may facilitate transmission of arboviruses of blackflies through co-feeding and by modulating the host immune response [64, 66].

Even though the proteomic analysis included in the present study did reveal the presence of the certain well-known salivary proteins, proper identification of the proteins that were only immunogenic in ELISA-positive individuals appeared challenging with the currently available sialotranscriptomes. Furthermore, the identified proteins overall had a low number of matching peptides, which impedes drawing clear conclusions on immunogen characterization. This work thus highlights the importance of unravelling the sialotranscriptome of this important blackfly species as *S. damnosum* s.l. homologues of these and other salivary proteins might be important antigens.

Conclusions and further research avenues

Quantifying vector biting rates as well as measuring inter-individual variation in host exposure to vector bites, are crucial for parameterizing onchocerciasis transmission dynamics models and predicting the epidemiological impact of control and elimination interventions [6, 7]. The currently used *S. damnosum* s.l. HLC studies are usually conducted with a limited team of vector collectors who are positioned in few places known to have high fly densities, resulting in ABR values that are strongly biased towards measuring maximal biting rates rather than true exposure to vector bites. Both the IgG and IgM ELISA proposed in the present study allow for the measurement of antivector saliva antibody responses in humans and might be used to complement these *S. damnosum* s.l. HLC studies. Furthermore, with the exception of very few observational studies [11], interindividual variation in exposure has been modelled by applying a suitable statistical distribution to account for heterogeneous biting [7, 9]. Hence, quantifying the individuals' responses to simuliid salivary antigens and investigating its patterns with host age and sex would greatly assist datadriven calibration of onchocerciasis transmission models [7, 8].

Studies of mosquitoes and sand flies have shown clear seasonal fluctuations in anti-saliva antibody response as well as interpretable correlations with host exposure [15]. Since sampling for the present study was conducted during the (presumably) high biting season (August, wet season), further studies are now needed to evaluate the kinetics and longevity of responses with changes in biting rates (as a result of seasonal fluctuations [46] and anti-vectorial measures [13], and to investigate potential cross-reactions with other bloodsucking arthropods (such as biting midges, mosquitoes, and tabanids) in onchocerciasis-endemic areas. The proteomic results presented in this study provide valuable albeit preliminary information on *S. damnosum* s.l. salivary proteins. Further studies on the sialotranscriptome of *S. damnosum* s.l. are now needed to fully characterize the most immunogenic and specific proteins, which would ideally result in the production of recombinant antigens that can be employed in onchocerciasis epidemiological and vector control studies.

Supporting information

S1 File. Detailed description of assay optimization and additional results. Text A-Details of the IgG and IgM optimization process. Fig A-Optical density (OD) values for the preliminary

immunoassays. Fig B–Distribution of optical density (OD) values for optimization process of immunoassays. Fig C–Comparison of two negative control groups for validation of final IgG ELISA. Table A–Summary statistics of preliminary (OD) and final (SOD) IgG and IgM ELISA. (DOCX)

Acknowledgments

Special thanks go to the researchers at the Noguchi Memorial Institute for Medical Research in Accra, who allowed us access to their laboratory facilities enabling us to perform the first preliminary analyses. We gratefully acknowledge Pavel Talacko and Karel Harant from the Laboratory of Mass Spectrometry, Biocev, Charles University who performed the proteomic and mass spectrometric analyses.

Author Contributions

Conceptualization: Laura Willen, Maria-Gloria Basáñez, Orin Courtenay, Petr Volf.

Data curation: Laura Willen.

Formal analysis: Laura Willen, Vit Dvorak.

- Funding acquisition: Maria-Gloria Basáñez, Maha Elhadi Osman, Mike Y. Osei-Atweneboana, Orin Courtenay, Petr Volf.
- Investigation: Laura Willen, Vit Dvorak, Francis B. D. Veriegh, Frank T. Aboagye, Bright Idun.

Methodology: Laura Willen, Maria-Gloria Basáñez, Orin Courtenay, Petr Volf.

Project administration: Mike Y. Osei-Atweneboana, Orin Courtenay, Petr Volf.

Resources: Mike Y. Osei-Atweneboana, Petr Volf.

Supervision: Orin Courtenay, Petr Volf.

Validation: Laura Willen.

Visualization: Laura Willen.

Writing - original draft: Laura Willen.

Writing – review & editing: Laura Willen, Maria-Gloria Basáñez, Vit Dvorak, Francis B. D. Veriegh, Frank T. Aboagye, Bright Idun, Maha Elhadi Osman, Mike Y. Osei-Atweneboana, Orin Courtenay, Petr Volf.

References

- Colebunders R, Basáñez MG, Siling K, Post RJ, Rotsaert A, Mmbando B, et al. From river blindness control to elimination: bridge over troubled water. Infect Dis Poverty. 2018; 7: 1–15. <u>https://doi.org/10. 1186/s40249-017-0384-1</u> PMID: 29335021
- Adler PH, Cheke RA, Post RJ. Evolution, epidemiology, and population genetics of black flies (Diptera: Simuliidae). Infect Genet Evol. 2010; 10: 846–865. <u>https://doi.org/10.1016/j.meegid.2010.07.003</u> PMID: <u>20624485</u>
- Basáñez MG, Churcher TS, Grillet ME. Chapter 11 Onchocerca-Simulium interactions and the population and evolutionary biology of Onchocerca volvulus. Adv Parasitol. 2009; 68: 263–313. <u>https://doi.org/ 10.1016/S0065-308X(08)00611-8</u> PMID: <u>19289198</u>
- World Health Organization. Ending the neglect to attain the Sustainable Development Goals–A road map for neglected tropical diseases 2021–2030 [Internet]. Geneva: World Health Organization. 2020. Available: <u>https://www.who.int/neglected_diseases/Ending-the-neglect-to-attain-the-SDGs—NTD-Roadmap.pdf?ua=1</u>.

- NTD Modelling Consortium Onchocerciasis Group. The World Health Organization 2030 goals for onchocerciasis: Insights and perspectives from mathematical modelling. Gates Open Res. 2019; 3: 1– 16. https://doi.org/10.12688/gatesopenres.12862.3 PMID: 31259312
- Walker M, Stolk WA, Dixon MA, Bottomley C, Diawara L, Traoré MO, et al. Modelling the elimination of river blindness using long-term epidemiological and programmatic data from Mali and Senegal. Epidemics. 2017; 18: 4–15. https://doi.org/10.1016/j.epidem.2017.02.005 PMID: 28279455
- Hamley JID, Milton P, Walker M, Basáñez MG. Modelling exposure heterogeneity and density dependence in onchocerciasis using a novel individual-based transmission model, EPIONCHO-IBM: Implications for elimination and data needs. PLoS Negl Trop Dis. 2019; 13: 1–21. <u>https://doi.org/10.1371/journal.pntd.0007557</u> PMID: <u>31805049</u>
- Hamley JID, Walker M, Coffeng LE, Milton P, de Vlas SJ, Stolk WA, et al. Structural Uncertainty in Onchocerciasis Transmission Models Influences the Estimation of Elimination Thresholds and Selection of Age Groups for Seromonitoring. J Infect Dis. 2020; 221: S510–S518. <u>https://doi.org/10.1093/ infdis/jiz674</u> PMID: <u>32173745</u>
- Winnen M, Plaisier AP, Alley ES, Nagelkerke NJD, Van Oortmarssen G, Boatin BA, et al. Can ivermectin mass treatments eliminate onchocerciasis in Africa? Bull World Health Organ. 2002; 80: 384–390. https://doi.org/10.1590/S0042-96862002000500009 PMID: 12077614
- Walsh JF, Davies JB, Le Berre R, Garms R. Standardization of criteria for assessing the effect of *Simulium* control in onchocerciasis control programmes. Trans R Soc Trop Med Hyg. 1978; 72: 675–676. https://doi.org/10.1016/0035-9203(78)90039-1 PMID: 734734
- 11. Jacobi CA, Enyong P, Renz A. Individual exposure to *Simulium* bites and intensity of *Onchocerca volvulus* infection. Parasit Vectors. 2010; 3: 53. https://doi.org/10.1186/1756-3305-3-53 PMID: 20565835
- Routledge I, Walker M, Cheke RA, Bhatt S, Nkot PB, Matthews GA, et al. Modelling the impact of larviciding on the population dynamics and biting rates of *Simulium damnosum* (s.l.): Implications for vector control as a complementary strategy for onchocerciasis elimination in Africa. Parasit Vectors. 2018; 11: 1–16. <u>https://doi.org/10.1186/s13071-017-2573-y</u> PMID: 29291748
- Jacob BG, Loum D, Lakwo TL, Katholi CR, Habomugisha P, Byamukama E, et al. Community-directed vector control to supplement mass drug distribution for onchocerciasis elimination in the Madi mid-North focus of Northern Uganda. PLoS Negl Trop Dis. 2018; 12. <u>https://doi.org/10.1371/journal.pntd.</u> 0006702 PMID: <u>30148838</u>
- Loum D, Cozart D, Lakwo T, Habomugisha P, Jacob B, Cupp EW, et al. Optimization and evaluation of the Esperanza Window Trap to reduce biting rates of *Simulium damnosum* sensu lato in Northern Uganda. PLoS Negl Trop Dis. 2019; 13: 1–17. <u>https://doi.org/10.1371/journal.pntd.0007558</u> PMID: <u>31310607</u>
- Quinnell RJ, Soremekun S, Bates PA, Rogers ME, Garcez LM, Courtenay O. Antibody response to sand fly saliva is a marker of transmission intensity but not disease progression in dogs naturally infected with *Leishmania infantum*. Parasit Vectors. 2018; 11: 1–12. <u>https://doi.org/10.1186/s13071-017-2573-y PMID: 29291748</u>
- Clements MF, Gidwani K, Kumar R, Hostomska J, Dinesh DS, Kumar V, et al. Measurement of recent exposure to *Phlebotomus argentipes*, the vector of indian visceral leishmaniasis, by using human antibody responses to sand fly saliva. Am J Trop Med Hyg. 2010; 82: 801–807. <u>https://doi.org/10.4269/ ajtmh.2010.09-0336</u> PMID: <u>20439958</u>
- Ribeiro JMC, Francischetti IMB. Role of arthropod saliva in blood feeding: sialome and post-sialome perspectives. Annu Rev Entomol. 2003; 48: 73–88. <u>https://doi.org/10.1146/annurev.ento.48.060402.102812</u> PMID: <u>12194906</u>
- Drame PM, Poinsignon A, Besnard P, Le Mire J, Dos-Santos MA, Sow CS, et al. Human antibody response to Anopheles gambiae saliva: An immuno-epidemiological biomarker to evaluate the efficacy of insecticide-treated nets in malaria vector control. Am J Trop Med Hyg. 2010; 83: 115–121. <u>https://doi.org/10.4269/ajtmh.2010.09-0684</u> PMID: 20595489
- Dama E, Cornelie S, Bienvenu Somda M, Camara M, Kambire R, Courtin F, et al. Identification of *Glossina palpalis gambiensis* specific salivary antigens: Towards the development of a serologic biomarker of human exposure to tsetse flies in West Africa. Microbes Infect. 2013; 15: 416–427. <u>https://doi.org/10.1016/j.micinf.2013.03.001</u> PMID: 23500186
- Schwarz A, Helling S, Collin N, Teixeira CR, Medrano-Mercado N, Hume JCC, et al. Immunogenic salivary proteins of *Triatoma infestans*: Development of a recombinant antigen for the detection of low-level infestation of triatomines. PLoS Negl Trop Dis. 2009; 3. <u>https://doi.org/10.1371/journal.pntd.0000532</u> PMID: 19841746
- 21. Willen L, Lestinova T, Kalouskova B, Sumova P, Spitzova T, Velez R, et al. Field study of the improved rapid sand fly exposure test in areas endemic for canine leishmaniasis. PLoS Negl Trop Dis. 2019; 13: e0007832. https://doi.org/10.1371/journal.pntd.0007832 PMID: <u>31751334</u>

- Cross ML, Cupp MS, Cupp EW, Galloway AL, Enriquez FJ. Modulation of murine immunological responses by salivary gland extract of *Simulium vittatum* (Diptera: Simuliidae). J Med Entomol. 1993; 30: 928–935. <u>https://doi.org/10.1093/jmedent/30.5.928</u> PMID: 8254643
- Cross ML, Cupp MS, Cupp EW, Ramberg FB, Enriquez FJ. Antibody Responses of BALB/c mice to salivary antigens of hematophagous black flies (Diptera: Simuliidae). J Med Entomol. 1993; 30: 725–734. https://doi.org/10.1093/jmedent/30.4.725 PMID: 8360895
- Hellberg W, Mellor PS, Torsteinsdottir S, Marti E. Insect bite hypersensitivity in the horse: Comparison of IgE-binding proteins in salivary gland extracts from *Simulium vittatum* and *Culicoides nubeculosus*. Vet Immunol Immunopathol. 2009; 132: 62–67. <u>https://doi.org/10.1016/j.vetimm.2009.09.016</u> PMID: <u>19836841</u>
- Hempolchom C, Sookrung N, Srisuka W, Reamtong O, Sakolvaree Y, Chaicumpa W, et al. Characterization of IgE-binding proteins in the salivary glands of *Simulium nigrogilvum* (Diptera: Simuliidae). Parasitol Res. 2019; 118: 2353–2359. <u>https://doi.org/10.1007/s00436-019-06383-x</u> PMID: <u>31263951</u>
- Hougard J-M, Alley ES, Yaméogo L, Dadzie KY, Boatin BA. Eliminating onchocerciasis after 14 Years of vector control: A proved strategy. J Infect Dis. 2001; 184: 497–503. <u>https://doi.org/10.1086/322789</u> PMID: <u>11471108</u>
- Alley ES, Plaisier AP, Boatin BA, Dadzie KY, Remme J, Zerbo G, et al. The impact of five years of annual ivermectin treatment on skin microfilarial loads in the onchocerciasis focus of Asubende, Ghana. Trans R Soc Trop Med Hyg. 1994; 88: 581–584. <u>https://doi.org/10.1016/0035-9203(94)90172-4</u> PMID: 7992347
- Post RJ, Cheke RA, Boakye DA, Wilson MD, Osei-Atweneboana MY, Tetteh-Kumah A, et al. Stability and change in the distribution of cytospecies of the *Simulium damnosum* complex (Diptera: Simuliidae) in southern Ghana from 1971 to 2011. Parasit Vectors. 2013; 6: 1–11. <u>https://doi.org/10.1186/1756-</u> 3305-6-1 PMID: 23281838
- Lamberton PHL, Cheke RA, Walker M, Winskill P, Osei-Atweneboana MY, Tirados I, et al. Onchocerciasis transmission in Ghana: Biting and parous rates of host-seeking sibling species of the *Simulium damnosum* complex. Parasit Vectors. 2014; 7:511. <u>https://doi.org/10.1186/s13071-014-0511-9</u> PMID: 25413569
- Lamberton PHL, Cheke RA, Winskill P, Tirados I, Walker M, Osei-Atweneboana MY, et al. Onchocerciasis transmission in Ghana: Persistence under different control strategies and the role of the simuliid vectors. PLoS Negl Trop Dis. 2015; 9: 1–27. <u>https://doi.org/10.1371/journal.pntd.0003688</u> PMID: 25897492
- Cheke RA, Basanez M, Perry M, White MT, Garms R, Obuobie E, et al. Potential effects of warmer worms and vectors on onchocerciasis transmission in West Africa. Philos Trans R Soc B. 2015; 370: 20130559. <u>https://doi.org/10.1098/rstb.2013.0559</u> PMID: <u>25688018</u>
- **32.** Plaisier AP, van Oortmarssen GJ, Habbema JDF, Remme J, Alley ES. ONCHOSIM: a model and computer simulation program for the transmission and control of onchocerciasis. Comput Methods Programs Biomed. 1990; 31: 43–56. <u>https://doi.org/10.1016/0169-2607(90)90030-d</u> PMID: 2311368
- Basáñez MG, Walker M, Turner HC, Coffeng LE, de Vlas SJ, Stolk WA. River Blindness: Mathematical Models for Control and Elimination. Adv Parasitol. 2016; 94: 247–341. <u>https://doi.org/10.1016/bs.apar.</u> 2016.08.003 PMID: <u>27756456</u>
- 34. Frempong KK, Walker M, Cheke RA, Tetevi EJ, Gyan ET, Owusu EO, et al. Does increasing treatment frequency address suboptimal responses to ivermectin for the control and elimination of river blindness? Clin Infect Dis. 2016; 62: 1338–1347. <u>https://doi.org/10.1093/cid/ciw144</u> PMID: <u>27001801</u>
- **35.** Turner HC, Osei-Atweneboana MY, Walker M, Tettevi EJ, Churcher TS, Asiedu O, et al. The cost of annual versus biannual community-directed treatment of onchocerciasis with ivermectin: Ghana as a case study. PLoS Negl Trop Dis. 2013; 7: e2452. <u>https://doi.org/10.1371/journal.pntd.0002452</u> PMID: 24069497
- 36. Sumova P, Sima M, Spitzova T, Osman ME, Guimaraes-Costa AB, Oliveira F, et al. Human antibody reaction against recombinant salivary proteins of *Phlebotomus orientalis* in Eastern Africa. PLoS Negl Trop Dis. 2018; 12: e0006981. https://doi.org/10.1371/journal.pntd.0006981 PMID: 30513081
- 37. Veriegh FBD. Impacts of semi-annual ivermectin treatment and the spatio-temporal distribution and abundance of *Simulium damnosum* sensu lato vectors on the transmission of *Onchocerca volvulus*. PhD thesis, Kwame Nkrumah University of Science and Technology (KNUST). 2020.
- Tang J, Toe L, Back C, Zimmerman P, Pruess K, Unnasch T. The Simulium damnosum species complex: phylogenetic analysis and molecular identification based upon mitochondrially encoded gene sequences. Insect Mol Biol. 1995; 4: 79–88. <u>https://doi.org/10.1111/j.1365-2583.1995.tb00011.x</u> PMID: 7551196
- Hall T. BioEdit: a user-friendly biological sequence alignment editor and analysis program for Windows 95/98/NT. Nucleic Acids Symposium Series. 1999; 41: 95–98.

- R Core Team (2015). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. [Internet]. Available: <u>https://www.r-project.org/</u>
- 41. Wickham H. ggplot2: Elegant Graphics for Data Analysis. Springer-Verlag New York, 2009.
- Hebert AS, Richards AL, Bailey DJ, Ulbrich A, Coughlin EE, Westphall MS, et al. The one hour yeast proteome. Molecular and Cellular Proteomics. 2014; 13: 339–347. <u>https://doi.org/10.1074/mcp.M113. 034769</u> PMID: <u>24143002</u>
- Cox J, Hein MY, Luber CA, Paron I, Nagaraj N, Mann M. Accurate proteome-wide label-free quantification by delayed normalization and maximal peptide ratio extraction, termed MaxLFQ. Mol Cell Proteomics. 2014; 13: 2513–2526. <u>https://doi.org/10.1074/mcp.M113.031591</u> PMID: 24942700
- Tyanova S, Temu T, Sinitcyn P, Carlson A, Hein MY, Geiger T, et al. The Perseus computational platform for comprehensive analysis of (prote)omics data. Nat Methods. 2016; 13: 731–740. <u>https://doi.org/</u> 10.1038/nmeth.3901 PMID: 27348712
- Oforka L, Adeleke M, Anikwe J, Hardy N, Mathias D, Makanjuola W, et al. Poor genetic differentiation but clear cytoform divergence among cryptic species in *Simulium damnosum* complex (Diptera: Simuliidae). Syst Entomol. 2018; 43: 123–135. https://doi.org/10.1111/syen.12256
- 46. Kostalova T, Lestinova T, Sumova P, Vlkova M, Rohousova I, Berriatua E, et al. Canine antibodies against salivary recombinant proteins of *Phlebotomus perniciosus*: A longitudinal study in an endemic focus of canine leishmaniasis. PLoS Negl Trop Dis. 2015; 9: e0003855. <u>https://doi.org/10.1371/journal.pntd.0003855</u> PMID: 26111018
- Elanga Ndille E, Doucoure S, Damien G, Mouchet F, Drame PM, Cornelie S, et al. First attempt to validate human IgG antibody response to Nterm-34kDa salivary peptide as biomarker for evaluating exposure to *Aedes aegypti* bites. PLoS Negl Trop Dis. 2012; 6: e1905. <u>https://doi.org/10.1371/journal.pntd.</u> 0001905 PMID: 23166852
- Orlandi-Pradines E, Almeras L, Denis de Senneville L, Barbe S, Remoué F, Villard C, et al. Antibody response against saliva antigens of *Anopheles gambiae* and *Aedes aegypti* in travellers in tropical Africa. Microbes Infect. 2007; 9: 1454–1462. <u>https://doi.org/10.1016/j.micinf.2007.07.012</u> PMID: <u>17913537</u>
- 49. Dorňáková V, Salazar-Sanchez R, Borrini-Mayori K, Carrion-Navarro O, Levy MZ, Schaub GA, et al. Characterization of guinea pig antibody responses to salivary proteins of *Triatoma infestans* for the development of a triatomine exposure marker. PLoS Negl Trop Dis. 2014; 8: e2783. <u>https://doi.org/10. 1371/journal.pntd.0002783 PMID: 24699441</u>
- Schaffartzik A, Marti E, Crameri R, Rhyner C. Cloning, production and characterization of antigen 5 like proteins from *Simulium vittatum* and *Culicoides nubeculosus*, the first cross-reactive allergen associated with equine insect bite hypersensitivity. Vet Immunol Immunopathol. 2010; 137: 76–83. <u>https://doi.org/10.1016/j.vetimm.2010.04.012 PMID: 20537727</u>
- Lestinova T, Vlkova M, Votypka J, Volf P, Rohousova I. *Phlebotomus papatasi* exposure cross-protects mice against *Leishmania major* co-inoculated with *Phlebotomus duboscqi* salivary gland homogenate. Acta Trop. 2015; 144: 9–18. <u>https://doi.org/10.1016/j.actatropica.2015.01.005</u> PMID: <u>25597641</u>
- 52. Rizzo C, Ronca R, Fiorentino G, Mangano VD, Sirima SB, Nèbiè I, et al. Wide cross-reactivity between Anopheles gambiae and Anopheles funestus SG6 salivary proteins supports exploitation of gSG6 as a marker of human exposure to major malaria vectors in tropical Africa. Malar J. 2011; 10: 1–10. <u>https:// doi.org/10.1186/1475-2875-10-1</u> PMID: <u>21214892</u>
- Andersen JF, Pham M Van, Meng Z, Champagne DE, Ribeiro JMC. Insight into the sialome of the black fly, *Simulium vittatum*. J Proteome Res. 2009; 8: 1474–1488. <u>https://doi.org/10.1021/pr8008429</u> PMID: <u>19166301</u>
- Botto C, Villamizar NJ, Jokić Z, Noya-Alarcón O, Cortés J, Escalona M, et al. Landscape epidemiology of human onchocerciasis in southern venezuela. In: Nriagu J(Ed). Encyclopedia of Environmental Health. 2019; 13–32 (Elsevier). https://doi.org/10.1016/B978-0-12-409548-9.01790–5
- 55. Ribeiro JMC, Valenzuela JG, Pham VM, Kleeman L, Barbian KD, Favreau AJ, et al. An insight into the sialotranscriptome of *Simulium nigrimanum*, a black fly associated with fogo selvagem in South America. Am J Trop Med Hyg. 2010; 82: 1060–1075. <u>https://doi.org/10.4269/ajtmh.2010.09-0769</u> PMID: 20519601
- 56. Chagas AC, Calvo E, Pimenta PFP, Ribeiro JMC. An insight into the sialome of *Simulium guianense* (Diptera: Simuliidae), the main vector of river blindness disease in Brazil. BMC Genomics. 2011; 12: 612. <u>https://doi.org/10.1186/1471-2164-12-612</u> PMID: <u>22182526</u>
- 57. Mattah PAD, Futagbi G, Amekudzi LK, Mattah MM, De Souza DK, Kartey-Attipoe WD, et al. Diversity in breeding sites and distribution of *Anopheles* mosquitoes in selected urban areas of southern Ghana. Parasit Vectors. 2017; 10: 25. <u>https://doi.org/10.1186/s13071-016-1941-3</u> PMID: <u>28086941</u>
- 58. Schaffartzik A, Weichel M, Crameri R, Björnsdóttir TS, Prisi C, Rhyner C, et al. Cloning of IgE-binding proteins from *Simulium vittatum* and their potential significance as allergens for equine insect bite

hypersensitivity. Vet Immunol Immunopathol. 2009; 132: 68–77. <u>https://doi.org/10.1016/j.vetimm.2009.</u> 09.017 PMID: <u>19836085</u>

- 59. Cupp M, Cupp E, Ochoa-a J, Moulton J. Salivary apyrase in New World blackflies (Diptera: Simuliidae) and its relationship to onchocerciasis vector status. Med Vet Entomol. 1995; 9: 325–330. <u>https://doi.org/ 10.1111/j.1365-2915.1995.tb00141.x</u> PMID: <u>7548952</u>
- 60. Cupp MS, Ramberg FB, Cupp EW. Salivary apyrase in African and New World vectors of *Plasmodium* species and its relationship to malaria transmission. Am J Trop Med Hyg. 1994; 50: 235–240. <u>https://doi.org/10.4269/ajtmh.1994.50.235</u> PMID: <u>8116818</u>
- Peng Z, Simons FER. Mosquito allergy: Immune mechanisms and recombinant salivary allergens. Int Arch Allergy Immunol. 2004; 133: 198–209. <u>https://doi.org/10.1159/000076787</u> PMID: <u>14963366</u>
- Volfova V, Tothova V, Volf P. Hyaluronidase activity in the salivary glands of tabanid flies. Insect Biochem Mol Biol. 2016; 73: 38–46. <u>https://doi.org/10.1016/j.ibmb.2016.03.007</u> PMID: <u>27045753</u>
- 63. Rádrová J, Vlková M, Volfová V, Sumová P, Cêtre-Sossah C, Carpenter S, et al. Hyaluronidase activity in saliva of European *Culicoides* (Diptera: Ceratopogonidae). J Med Entomol. 2016; 53: 212–216. <u>https://doi.org/10.1093/jme/tjv147</u> PMID: <u>26487248</u>
- 64. Ribeiro JMC, Charlab R, Rowton ED. Simulium vittatum (Diptera: Simuliidae) and Lutzomyia longipalpis (Diptera: Psychodidae) salivary gland hyaluronidase activity. J Med Entomol. 2000; 37: 743–747. https://doi.org/10.1603/0022-2585-37.5.743 PMID: 11004788
- Volfova V, Hostomska J, Cerny M, Votypka J, Volf P. Hyaluronidase of bloodsucking insects and its enhancing effect on *Leishmania* infection in mice. PLoS Negl Trop Dis. 2008; 2: e294. <u>https://doi.org/ 10.1371/journal.pntd.0000294</u> PMID: <u>18820742</u>
- 66. Mead DG, Ramberg FB, Besselsen DG, Maré CJ. Transmission of vesicular stomatitis virus from infected to noninfected black flies co-feeding on nonviremic deer mice. Science. 2000; 287: 485–487. <u>https://doi.org/10.1126/science.287.5452.485</u> PMID: <u>10642550</u>